Author's response to reviews

Title: Leptomeningeal metastasis from hepatocellular carcinoma with other unusual metastases: a case report

Authors:

Zhenyu Pan (dr-zypan@163.com)
Guozi Yang (guoziyang_1982@163.com)
Tingting Yuan (31400135@qq.com)
Xiaochuan Pang (498917463@qq.com)
Yongxiang Wang (JLUWYX@sina.com)
Limei Qu (51331628@qq.com)
Lihua Dong (drlhdong@163.com)

Version: 4
Date: 28 April 2014

Author's response to reviews: see over
Dr. Yasunaru Sakuma  
Associate editor, *BMC Cancer*

Re: Manuscript “Leptomeningeal metastasis from hepatocellular carcinoma with other unusual metastases: a case report”. (MS: 4967101741197004)

Deer Dr. Sakuma,

Thank you for giving us the opportunity to further revise the manuscript. We have made extensive modifications to the revised manuscript in accordance with the editors’ suggestions. Please find attached the revised manuscript in which major changes have been emphasized by red font for easy review. A list of responses to every question from the editors is also enclosed. We hope that the revised manuscript is acceptable for publication.

I anticipate hearing from you soon.

With best wishes,  
Yours sincerely,

Lihua Dong  

Telephone: +86-158-43073216  
Fax: +86-431-8878-6172  
Email: drlhdong@163.com
Replies to editors
1. Please provide and put new MRI finding in your manuscript. Lesions in the right side of the cerebellopontine angle, indicating cranial nerve injury. This should become one of key finding.
   Response: The new MRI of LM was added in Figure 2.
2. Figure legend should be also changed in new one.
   Response: The figure legend was also revised.
3. The first case is very important to impress all audience. Your response comment is meaningful to recognize how you consider that this is LM metastases. Therefore, put and arrange your comment in the discussion.
   Response: Some comments were added in the Discussion to support our findings.